Experimental rapid and small-scale ecological population divergence in the absence of current natural selection

Gabriel Munar-Delgado¹*, Paula Hidalgo-Rodríguez², Gregorio Sánchez-Montes³, Jaime Potti⁴, José L. Tella⁵, Wolfgang Forstmeier⁶, Pim Edelaar²*

¹Department of Biodiversity, Ecology and Evolution, Faculty of Biological Sciences, Complutense University of Madrid, Madrid, Spain.

² Department of Molecular Biology and Biochemical Engineering, University Pablo de Olavide, Seville, Spain.

³ Department of Biodiversity and Evolutionary Biology, National Museum of Natural Sciences, CSIC, Madrid, Spain.

⁴ Department of Conservation Biology, Doñana Biological Station CSIC, Sevilla, Spain.

⁵ Department of Evolutionary Ecology, Doñana Biological Station CSIC, Sevilla, Spain.

⁶ Department of Ornithology, Max-Planck-Institut für biologische intelligenz, Seewiesen, Germany.

Corresponding author: gabriel.munar@gmail.com (GMD) and edelaar@upo.es (PE).

Ecological divergence without natural selection

Key words: adaptation, assortative mating, ecological divergence, habitat choice, matching habitat choice, natural selection

Abstract

Adaptive divergence has long been a core topic in the field of evolutionary biology, with natural selection traditionally considered its only driver. Here we focus on the ability of matching habitat choice to generate population divergence and reproductive isolation. This alternative mechanism of divergence considers that individuals choose their habitats based on an evaluation of the ecological match between their phenotype and the available environments, which subsequently limits gene flow. To test this, we conducted experiments with captive zebra fiches equipped with transponder-tags and using transponder-operated bird feeders. We thereby created within a single aviary two areas with distinct resources, and the associated ecological traits that provided access to only one of the resources. We found that most zebra finches chose to breed in the same area as where they had access to their ecological resource, thereby creating population divergence in the absence of current natural selection on the ecological trait. This choice of breeding area indirectly resulted in assortative mating for the ecological trait. If the ecological trait were heritable, this assortative mating would carry the obtained divergence into the next generation. Our results experimentally confirm the predicted ability of matching habitat choice to drive rapid population divergence and limit maladaptive gene flow, especially at the small temporal and spatial scales where natural selection is unlikely to do so (here: one generation and one aviary). This might be increasingly relevant in a world where anthropogenic impacts create rapid environmental changes.

Lay summary

Species face unprecedented environmental challenges due to human activities rapidly altering habitats and biological communities. Understanding how species adapt is therefore crucial for conservation of endangered species and management of invasive species. Natural selection, the cornerstone of adaptive evolution, has long been considered the only driver of adaptation to the environment. Our study delves into an intriguing alternative driver called Matching Habitat Choice (MHC), where animals select their environments based on their evaluation of how well their traits match ecologically with the available local environments. Using captive zebra finches in an aviary, our research examined how these birds make choices about where to settle and breed in relation to locally available food resources. We observed that these birds preferred breeding sites matching their ecological traits. We therefore confirmed that population divergence is possible without current natural selection. Our findings carry significant implications for biodiversity management amidst rapid environmental changes. They underscore the power of individual decision-making in shaping evolutionary dynamics, offering insights for ways to preserve species in an ever-changing world.

Introduction

Immersed in the Anthropocene, species are experiencing a period of unprecedented novel and rapid environmental changes (Waters et al., 2016) leading to an enormous loss of biodiversity, with extinction rates far above pre-human levels (Johnson et al., 2017). Human activity also results in the spread of invasive alien species, which impact biodiversity, economy and health (Pyšek et al., 2020). In this context of environmental change, one of the biggest challenges in evolutionary ecology is understanding in detail how species and populations adapt to their environment, to improve management and conservation strategies for native species as well as control plans for invasive species (Garant, 2020; Otto, 2018). Experiments on adaptation began more than 70 years ago and have provided much insight (Rice & Hostert, 1993; Schlichting, 2021; Wadgymar et al., 2022). However, research on adaptation has mostly focused on natural selection and phenotypic plasticity, while other mechanisms that could result in the same outcomes have been relatively neglected (Edelaar & Bolnick, 2019; Trappes et al., 2022). This could be biasing our knowledge and, consequently, our capacity for action.

One alternative mechanism for adaptation is Matching Habitat Choice (MHC). MHC is an ecological process based on individuals' self-assessment of local performance across habitats, followed by settling (or spending more time) where performance is highest (Edelaar et al., 2008). MHC is thereby responsive to an individual's ecological performance and therefore to variation in phenotypes. By doing so, MHC has been hypothesized to influence a large number of ecological and evolutionary phenomena, such as individual and population fitness, local adaptation, maintenance of genetic variation, ecological population divergence, positive assortative mating and even speciation (Berner & Thibert-Plante, 2015; Edelaar et al., 2008; Nicolaus & Edelaar,

2018; Porter & Akcali, 2020; Porter & Benkman, 2022; Ravigné et al., 2009; Scheiner et al., 2022).

Despite the potential ecological and evolutionary implications of MHC and a noticeable rise in interest in it in recent years, the empirical evidence supporting it is still scarce, in part because it is overlooked, in part because it is hard to obtain convincing support that excludes alternative interpretations (Edelaar et al., 2019; Edelaar & Bolnick, 2012). Because of this, MHC has only been confirmed to influence a few of the phenomena it has been predicted to cause. Specifically, there has not been any experimental test for whether MHC can drive assortative mating and reproductive isolation (but see Porter & Benkman, 2022 for an observational study). This could arise because the phenotype-environment covariance due to MHC could subsequently result in individuals mating with other individuals with a similar phenotype, if mating occurs within the chosen habitat. This is important, as local mating then maintains genetic variation at the metapopulation level, translates into reduced gene flow between individuals with different phenotypes, and could provide an initial step towards speciation (Berner & Thibert-Plante, 2015; Bolnick & Otto, 2013; Edelaar et al., 2019; Nicolaus & Edelaar, 2018). Basically, assortative mating due to MHC means that the populationstructuring effect of MHC is not lost during reproduction, and propels this effect into the next generation. A test of whether MHC can cause assortative mating is therefore long overdue.

To test whether MHC could result in population divergence and assortative mating we designed an experiment that allowed us to manipulate local performance (here: individual food intake rate) across local environments. We created a spatially-structured environment by placing two sets of transponder-operated feeders at the two extremes of an aviary. We manipulated individual variation in local performance by marking a population of captive zebra finches (*Taeniopygia guttata*) with a leg band containing a built-in transponder tag. These transponders allowed half of the individuals to have access to food only at the feeders placed in one area of the aviary (area A), and the other half of the individuals to have access only in the other area (area B). Thus, with the electronic feeders we mimicked the existence of two types of local habitats differing in resources (areas A and B), and with the transponder tags (types A and B) we mimicked the existence of two ecological traits differing in providing access to these resources, each matching only one of the two available resources. As an example of something similar occurring in

nature one could think of crossbills (*Loxia curvirostra* complex). These birds feed on seeds from conifer cones and have different bill morphologies (the ecological trait) that match cone morphologies of different tree species (the local habitat), and crossbills are thought to disperse across the mosaic of patches of different conifer species to increase this match and thereby their food intake rate (Benkman, 2017; Porter & Benkman, 2022). In this case, we provided a novel environment where birds had to locally evaluate a familiar ecological performance (food intake rate). By providing nest boxes at both areas, we let individuals assess their local performance in both areas and choose in which area to breed. We determine where each individual bred and with whom through behavioural observations. At the end of the experiment, all adults and fledglings were genotyped to assess genetic parentage, allowing us to associate adults with the nests that contained their fledglings at the genetic level.

For theoretical reasons, we predicted that in this experimental setup: (1) individuals will breed in the same area where they have access to food; (2) individuals thereby pair with other individuals with the same transponder type; (3) pairs will be producing more fledglings if they breed in the area where they have access to food; (4) MHC will be stronger at the genetic level, because it will promote extra-pair copulations with individuals that have access to food in the same area; (5) extra-pair copulations will be assortative; (6) produced offspring would be locally matching.

Materials and Methods

Experimental design

We performed two experiments. In experiment 1 we used 70 zebra finches obtained from different providers. In experiment 2, we used 130 zebra finches raised in our experimental aviary (offspring of individuals different from those in experiment 1). Birds were uniquely colour-banded and equipped with a leg-band with a passive integrated transponder (PIT)-tag (Eccel Technology Ltd.) for visual and electronic identification.

The experimental indoor aviary was divided into four interconnected areas delimited by plastic mesh (Fig. 1). Several mosquito net strips were placed in the two intermediate areas to increase the difficulty of flying between the extreme areas. These extreme areas, designated as breeding areas A and B, were equipped with perches and water stations throughout the entire period, encompassing both acclimation and experimental phases. Additionally, feeders and nest boxes were positioned within the breeding areas during the experiment (Fig. 1).



Figure 1. Schematic representation of the experimental aviary. A and B denote the name of each breeding area.

We used 18 electronically operated feeders (NatureCounters) able to read and respond to PIT-tags to manipulate individuals' local ecological performance. These feeders where usually closed but gave access to seeds when a specific birds with a specific PIT-tag perched on them.

After the acclimatization phase (Supplementary material), we retained 28 males and 28 females for experiment 1 and 38 males and 32 females for experiment 2 (Supplementary material). Subsequently, for both experiments, we placed half of the feeders in each breeding area (A and B). Following this, we randomly assigned half of the individuals to group A, providing them access to seeds only at feeders in area A, and the other half to group B, allowing them access to seeds only in area B. This method generated two distinct groups with differing ecological performances (food access) in both breeding areas. Next, we placed nest boxes in both breeding areas (28 in experiment 1 and 34 in experiment 2, half of them in each area) and let the individuals breed.

Nests were checked weekly, starting three weeks after placing the nest boxes, continuing until the last fledgling left its nest. Nestlings were banded at the age of 10 ± 3

days (value \pm range). Breeder identification was based on colour bands, direct visual observations, and camera recordings (network video recorder and 1080P camera, Sannce) (Supplementary material). We classified the breeding individuals as being part of a heterosexual pair, same-sex pair, or trio (when 3 individuals allowed each other to enter the nest). To terminate each experiment, we removed eggs laid after the first fledgling had left its nest (i.e., removed potential second clutches).

The experiments were performed in accredited installations (ES410910008004) and approved by the relevant authorities (Consejería de Agricultura, Pesca y Desarrollo Rural of the Junta de Andalucía, 28/03/2018/040).

Genotyping and parentage analysis

Blood samples were obtained from all adults and fledglings at the end of the experiments. DNA isolation was performed using an extraction robot (Freedom EVO 100, Tecan), and followed by genotyping at 10 microsatellite loci (Forstmeier, Schielzeth, et al., 2007, Table S9) using PCR (see Supplementary material for PCR conditions).

For parentage analysis, we utilized CERVUS version 3.0.7 (Marshall et al., 1998), employing a likelihood-based approach to determine parentage. Using specific simulation settings for each experiment (10,000 offspring, all adult individuals as candidate parents, 0.01 rate of loci mistyped), we assigned parents based on the highest log-likelihood ratio score (LOD) without specifying known parents. This allowed us to identify genetic parents of fledglings and associate each fledgling with two reproductive adults, linking each parent to the nest(s) and area(s) where its fledglings hatched.

Establishment of phenotype-environment matching

Breeding individuals (see criteria in Supplementary material) were classified as matching individuals if they had access to food in the same area where they chose to breed (i.e., a match between transponder type and breeding area). Individuals breeding in the opposite area to their feeding site were classified as non-matching.

Through the use of genetic parentage analysis, reproductive individual-nest combinations (see above) were classified as matching if the reproductive individual (i.e., genetic parent) had access to food in the same area where the nest was and as nonmatching if not. Consequently, at the genetic level, reproductive individuals could have been matching and non-matching at the same time if they had genetic offspring in nests in both areas.

We also classified fledglings as matching or not-matching. For this we assumed that offspring inherited the transponder type from their genetic parents in the following way. If both genetic parents had the same transponder type, the offspring inherited the transponder type of their parents. However, if genetic parents had different transponders, each offspring was assigned to inherit either transponder type with an equal probability. Finally, we compared the virtually inherited transponder type with the side of the aviary where offspring had fledged to classify them as "matching" or "non-matching".

Statistical analysis

All statistical analyses were conducted with R 3.4.1 (R Core Team, 2020). GLMs and GLMMs were fitted with functions glm and glmer, respectively, using the package 'lme4' (Bates et al., 2015). For all models, experiment number was included as a fixed effect to account for any differences between experiments (with only two levels, we decided not to fit it as a random effect). We don't explicitly investigate and discuss differences between experiments, which seem mild at most (suggesting our results are robust with respect to this variation in design). Model predictions and 95% confidence intervals (CI) were obtained with the function 'get_model_data' from the package 'sjPlot' (Lüdecke, 2022). Statistical significance of variables is based on log-likelihood ratio tests comparing models with versus without the tested variable.

Do individuals breed in the same area where they have access to food?

We tested for MHC using a binomial GLM with breeding area as the binary response variable, and transponder type (A/B) for both breeding males and females as the explanatory variable. We only included data from heterosexual pairs to be able to control for the effect of the opposite sex (trios and same-sex pairs turned out to be relatively common; see Supplementary material). To simplify the model, we excluded two pairs with a repeated (polygynous) male. To increase sample size, we also tested for MHC for the whole breeding population, including same-sex pairs and trios. We fitted breeder ID as a random effect (GLMM) to account for repeated measures for two polygynous males.

Fixed effects included transponder type of the breeding individual, sex, and their interaction to test for any differential effect of transponder type between the sexes.

For the models above, we only included unique breeding individual-nest combinations, so not taking into account the number of offspring hatched for each combination. We did this because breeding area selection could be affecting the number of offspring (e.g., having more offspring when breeding at the area where they had access to food), and we preferred to give equal weight to each individual breeder. Also, breeding area selection is not done independently for each offspring (they are typically produced in the same nest), so by focusing on unique individual-nest combinations we avoid inflated statistical significance due to pseudoreplication.

Do individuals pair with individuals with the same transponder type?

To test if the proportion of pairs that bred assortatively for transponder type (i.e., local ecological performance) differed from that expected if mating was random (50% assortative), we used a Chi-square test of expected frequencies. We only included heterosexual pairs to ensure that pairs had not already formed before the beginning of the experiment (Adkins-Regan & Krakauer, 2000). We did not include trios either because if there were two individuals with the same type of transponder and one with a different one, we could not determine if the trio was assortative or not.

Do pairs produce more fledglings if they breed in the area where they have access to food?

We investigated if any MHC for breeding had an effect on reproductive success (i.e., led to adaptation). For this, we tested whether the number of matching adults per breeding pair (0, 1 or 2) affected their reproductive success (GLM, with the number of fledglings as response variable and assuming a Poisson error distribution).

How strong is the matching breeding habitat choice at the genetic level?

We tested for matching breeding habitat choice at the genetic level (i.e., a match between the area where the genetic parent had access to food and where its fledglings hatched) using a binomial GLMM with the breeding area where their fledglings hatched as the response variable. We fitted a separate model for males and females. Fixed factor was individual transponder type (A/B). For both models, we included individual ID as a random effect to account for individuals with fledglings in different nest boxes. After observing the results for the males, we added extra-pair condition (whether a fledgling is raised in a focal nest due to an extra-pair fertilization) and its interaction with male transponder type as a fixed effect. To classify genetic males (fathers) as either within-pair or extra-pair mates, we only included males from heterosexual pairs since we could not be sure about breeding bonds with males in same-sex pairs with two females or in trios. We considered this interaction because the effect of male transponder type on breeding area appeared to be virtually absent at the genetic level, and we hypothesized that (in contrast to our initial prediction) the effect could be the opposite in extra-pair fertilizations (e.g., males have offspring with their breeding pair in the area where they feed, but have extra-pair offspring in the opposite area).

For the models above, we only included unique individual-nest combinations for the same reasons as for testing MHC for breeding individuals.

Are within-pair and extra-pair parentage assortative for transponder type?

After identifying extra-pair paternity (see above) we extracted each male-female genetic combination that resulted in within-pair and extra-pair offspring. Then, for both groups, we tested if the observed proportion of male-female genetic combinations that were assortative for transponder type differed from that expected if extra-pair mating was random (50% assortative). We used a Chi-square test of expected frequencies and, to avoid pseudoreplication, each unique combination was only counted once.

Do the (virtual) phenotypes of the produced fledglings and the entire population match the local environment?

Finally, to test if the observed proportion of matching fledglings out of all produced fledglings differed from that expected by chance (50% matching), we also used a Chi-square test of expected frequencies. The same test was used to test if the proportion of matching individuals for the entire population (breeding individuals plus fledglings) differed from that expected by chance (50% matching).

Results

As is usual for studies on zebra finches, not all individuals established pair bonds (Griffith et al., 1999). In addition, some individuals established pair bonds with an individual of the same sex, or with two individuals (Tables S1 and S2).

The majority of individuals bred in the same area where they had access to food

For heterosexual pairs (N=23), the probability for 9 males with transponder A to breed in area A was 0.85 [95% confidence interval: 0.38, 0.98], significantly higher than for 14 males with transponder B (0.34 [0.12, 0.65]; LRT: $\chi 2$ (1) = 4.016, p = 0.045). Similarly, for 11 females with transponder A the probability to breed in area A was 0.79 [0.40, 0.96], much higher than for 12 females with transponder B, although the effect was not quite significant (0.33 [0.10, 0.70], LRT: $\chi 2$ (1) = 3.32, p = 0.069; Fig. 2A; Table S3). For the whole population of breeding individuals (including same-sex pairs, trios, and polygynous birds, total N = 68) and controlling for sex and its interaction with transponder type, the probability for individuals with transponder A to breed in area A was 0.86 [0.45, 0.98], significantly higher than for individuals with transponder B (0.19 [0.04, 0.57], LRT: $\chi 2$ (1) = 24.01, p < 0.00001; Fig. 2B; Table S4). The interaction effect between transponder type and sex was not significant (LRT: $\chi 2$ (1) = 0.513, p = 0.474).



Figure 2. Matching breeding habitat choice as based on nest location. (A) Model predictions (binomial GLM) for the probability of breeding in area A for females (grey diamonds) and males (red diamonds) depending on their transponder type, where remaining covariates are set to their means. Error bars represent the 95% CI of model predictions. Dots are raw data for breeding females (grey) and males (red). (B) Model predictions (binomial GLMM) for all breeding individuals depending on their transponder type, where remaining covariates are set to their means. Error bars represent the 95% CI of model predictions. Brown dots are raw data.

The majority of individuals paired assortatively for transponder type

We found that 74% of heterosexual pairs (17 out of 23) were assortative for their transponder type ($\chi 2$ (1) = 5.26, p = 0.022) (Fig. 3).



Figure 3. Assortative mating for transponder type with respect to the social mate. Number of observed heterosexual disassortative or assortative breeding pairs (letters indicate the transponder type of the two pair members).

Individuals breeding in their matching area did not have greater reproductive success

For heterosexual pairs (N = 23), there was little evidence suggesting that matching individuals had higher reproductive success (LRT: $\chi 2$ (1) = 0.208, p = 0.648; Fig. 4; Table S5).



Figure 4. Reproductive success for breeding pairs. Number of fledglings for pairs with 0, 1 or 2 individuals whose transponder matches the breeding area. Black dots represent the model estimates and the error bars their 95% CI. Gray dots represent raw data.

The majority of females produced fledglings in their matching area but males produced them in both

At the genetic level, we identified a total of 34 reproductive female-nest combinations and 49 reproductive male-nest combinations. The probability of females with transponder A of having genetic fledglings in area A was 0.65 [0.36, 0.68] and for females with transponder B this was significantly lower at 0.16 [0.03, 0.58] (LRT: $\chi 2$ (1) = 8.07, p = 0.004; Fig. 5A; Table S6). However, males with transponder A and

transponder B showed no significant difference in their probability of having genetic fledglings in area A, at 0.56 [0.28, 0.81] and 0.34 [0.15, 0.61] respectively (LRT: $\chi 2$ (1) = 1.22, p = 0.268; Fig. 5A; Table S7). We tested if this smaller difference in males was because extra-pair offspring were predominantly produced in the area where they did not nest. Indeed, we observed a significant interaction between transponder type and within-pair/extra-pair male status (LRT: $\chi 2$ (1) = 4.32, p = 0.038; Fig. 5B; Table S8). For within-pair males, the probability for a male with transponder A or B of having offspring in area A was 0.81 [0.25, 0.98] versus 0.21 [0.03, 0.68] respectively. In contrast, the probability for an extra-pair male with transponder A or B of having fledglings in area A was closer to random (and, unexpectedly, even somewhat larger for males with transponder B), at 0.39 [0.09, 0.81] versus 0.55 [0.11, 0.93] respectively.



Figure 5. **Matching breeding habitat choice at the genetic level.** (**A**) Model predictions (binomial GLMM) for the probability of having genetic offspring in area A for females (grey diamonds) and males (red diamonds) depending on their transponder type. Remaining covariates are set to their means. Error bars represent the 95% CI of model predictions. Dots are raw data for females (grey) and males (red). (**B**) Model predictions (binomial GLMM) for the probability of having genetic offspring in area A for within-pair males (grey diamonds) and extra-pair males (brown diamonds) depending on their transponder type. Remaining covariates are set to their means. Error bars represent the 95% CI of model predictions (binomial GLMM) for the probability of having genetic offspring in area A for within-pair males (grey diamonds) and extra-pair males (brown diamonds) depending on their transponder type. Remaining covariates are set to their means. Error bars

represent the 95% CI of model predictions. Dots are raw data for within-pair males (grey) and extra-pair males (brown).

Genetic parentage was not assortative for transponder type

We found that 8 out of 16 (50%) unique genetic extra-pair combinations were assortative ($\chi 2$ (1) = 0.00, p = 1.00). The proportion of assortative combinations for within-pair parentage was higher (12 out of 18, 67%), but the difference between the number of assortative and disassortative combinations was not significant ($\chi 2$ (1) = 1.470, p = 0.225).

The majority of the fledgling population as well as the entire population were matching individuals

The number of locally matching fledglings (assuming virtual inheritance of parental transponders; see methods) was higher than expected by chance (53 out of 80, 66%) (χ 2 (1) = 8.45, p = 0.004). Globally, for the entire population (fledglings and breeding individuals, N = 148) there were more matching individuals than expected by chance (105 out of 148, 71%; χ 2 (1) = 25.97, p < 0.000001).

Discussion

Our study provides the first demonstration, to our knowledge, that MHC i) can drive ecological population divergence in a novel environment and be based on a novel trait, and ii) can result in local reproduction, thereby causing assortative mating, which maintains the divergence into the next generation. Together, this suggests that MHC can generally promote adaptive population divergence even in the absence of currently acting divergent natural selection. This possibility directly contradicts the widely held idea that only divergent natural selection can cause adaptive population divergence (e.g., reviewed in (Schluter, 2001; Wadgymar et al., 2022).

A priori, our predictions as listed in the introduction might have been considered to be unrealistic, since the flight distance between the two feeding areas was only a few meters. In the wild, Zebra finches habitually need to fly several kilometres between feeding and breeding areas (Zann et al., 1995), such that a few meters would hardly matter. It may also have seemed unrealistic to expect population divergence to arise in a virtually sympatric set up, when, in the wild, populations are genetically undifferentiated over hundreds of kilometres due to very high dispersal rates (Forstmeier, Segelbacher, et al., 2007; Zann et al., 1995). Nonetheless, in our experimental setting, zebra finches exposed to a manipulated differential food intake rate across habitats preferentially bred where their food intake rate was higher (although the effect did not quite reach statistical significance in the subset of heterosexual females, possibly because of a limited sample size). Or in other words, MHC arose and individuals had a differential habitat use based on their local ecological performance. This is not because the Zebra finches *could* not fly more (some actually bred in the non-matching area), but apparently because they did not *want* to fly more. In the wild, the trade-off between flight distance and food intake will often favour greater flight distances, and therefore reduced population divergence.

It could be argued that our experimental setting is very extreme, since zebra finches could feed at only one of the two areas. An alternative setup could have been to somehow set the feeders to open at different rates in both areas for both bird groups. However, this is not a qualitatively different design, it would only have reduced the magnitude of the effects of MHC. It is also important to note that birds were forced to feed in one of the two areas but they were not forced to breed there (and a few birds indeed chose to breed in their non-matching area). In this way, population divergence was not imposed by setting the feeding trade-off to the largest extent, and our design only maximized its probability of occurrence and its extent (i.e., the power of the experiment to avoid false negatives).

Our experimental results are largely consistent with predictions from theoretical (Bolnick & Otto, 2013; Edelaar et al., 2008, 2017, 2023; Edelaar & Bolnick, 2012; Mortier et al., 2019; Ravigné et al., 2009; Scheiner et al., 2022) and empirical studies (Benkman, 2017; Camacho et al., 2020; Camacho & Hendry, 2020; Edelaar et al., 2019; Holtmann et al., 2017; Porter & Benkman, 2022; Regan et al., 2022). Even more, MHC was able to operate in a novel environment and be based on a novel ecological trait. This corroborates that organisms can respond to novel ecological challenges via MHC. In other words, the pre-evolved mechanism of MHC can act in novel environments based on "basic" and generalizable aspects of individuals' ecological performance, such as food intake rate, without having to evolve de novo (Edelaar et al., 2008, 2019). This high

responsiveness of MHC in novel environments may be particularly relevant under rapid and novel human-induced habitat changes (Sih et al., 2011; Waters et al., 2016), and especially at small spatial and temporal scales where natural selection is unlikely to achieve divergence as effectively (Richardson et al., 2014). Thus, organisms could rapidly respond to new environmental challenges, even within a single generation, as long as the new local environments affect aspects of local performance that can somehow be evaluated by the organisms themselves.

We observed ecological population divergence, i.e., phenotype-environment correlation, due to MHC. Our experimental design allows us to exclude alternative explanations for its occurrence (Edelaar et al., 2019; Trappes et al., 2022). We can exclude imprinting or genetic habitat preference (Akcali & Porter, 2017; Edelaar et al., 2008) since the environments were novel, and individuals were assigned randomly to treatment groups. Phenotypic plasticity is excluded since individuals were not able to modify their transponder type. Adjustment of the environment (cf. Edelaar & Bolnick, 2019) is excluded since individuals were not able to modify (e.g., reprogram or re-engineer) the electronic feeders. Finally, we also prevented divergent natural (survival) selection from acting by not including a few individuals that were not able to learn how to use the electronic feeders (see Methods). This supports the view that individuals are not just *targets* of selection (i.e., actively select their environment) (Edelaar et al., 2019; Sultan et al., 2022).

Contrary to what we expected, we did not find a significant difference in reproductive success between matching and non-matching individuals (although our sample size is too small to detect weak effects). This might be because the travel distance between feeding areas was very short; longer distances are expected to have (higher) reproductive costs for non-matching individuals. The upshot of this lack of differential reproductive success is that divergent natural selection cannot explain the observed adaptive population divergence.

We confirmed that a non-random mating pattern emerged after manipulating local ecological performance. Most individuals bred with another individual with the same local performance as themselves, thus generating assortative mating for transponder type. This non-random mating could be an indirect effect of MHC, as predicted by other studies (Edelaar et al., 2008; Nicolaus & Edelaar, 2018; Porter & Akcali, 2020; Porter &

Benkman, 2022). If zebra finches spent most of the time in their matching area, then most encounters are expected to be between individuals with the same local performance. Therefore, pair bond formation (Maldonado-Chaparro et al., 2021) is also expected to occur between individuals in their matching area and with the same local performance. However, we cannot completely discard additional sources of assortative mating besides MHC. Zebra finches mate assortatively for traits such as neophobia (Pogány et al., 2018) and exploratory behaviour (Faust & Goldstein, 2021; Schuett et al., 2011), so it could be possible that they were also able to assess their potential partner's local performance and showed preferences based on that (Snowberg & Benkman, 2009). Because local performance is area specific, the inability of a bird (e.g., of type A) to feed in its non-matching area (B) could be negatively evaluated by a potential partner of the opposite local performance (B; and vice versa). However, zebra finches show no tendency to pair assortatively for overall phenotypic quality (including past reproductive performance; Wang et al., 2017), so perhaps this is a less likely occurrence.

Contrary to our prediction, transponder type did not affect extra-pair copulations, most likely due to the short distance between the two feeding areas as individuals had to fly only a few meters to change areas and look for extra-pair copulations. As a consequence, male zebra finches had offspring in both local habitats, independent of their own local performance. This could be due to male or female zebra finches looking for extra-pair copulations in their non-matching local habitat, e.g., to avoid mate guarding from their partners (Birkhead et al., 1988) which were feeding in their matching local habitat.

In spite of the frequency of extra-pair mating between birds belonging to opposite groups, if we consider the number of offspring produced by each male-female combination, the effect of MHC on population divergence (assuming virtual inheritance of the transponder type) was maintained in the offspring generations. The simulated inheritance of the transponder type showed that most of the fledged young would have hatched at their matching local habitat, the same habitat where they are expected to breed later as an adult. To the extent that MHC indirectly causes local reproduction, its effects on population structuring are thereby transmitted to the next generation. Hence, its withingeneration ecological consequences become between-generation evolutionary effects. In the hypothetical extreme case that individuals responded so strongly to the habitat variation that they all would only have reproduced in the area where they could feed, then all offspring would be expected to have identical ecological traits to their parents and to later also breed in the parental habitat. In effect then, we would have observed the formation of two ecologically specialized and reproductively fully isolated populations (i.e., biological species) in a single generation due to MHC. Indeed, in a similar experiment performed in the wild, we observed such a very high degree of reproductive isolation (Munar-Delgado et al., unpublished ms).

Despite the demonstrated potential relevance of MHC for adaptation, including to aspects of global change, and despite the fact that simulations suggest that in nature it might be as frequent as phenotypic plasticity (Nicolaus & Edelaar, 2018), there are few empirical studies that have tested for its existence in nature and have been able to rule out other processes (e.g., Camacho et al., 2020). To be able to better understand how individuals and populations adapt, and to improve biodiversity management under the scenario of global change, more research should focus on quantifying how important MHC is, and disentangle its potential contribution to divergence and adaptation in natural populations compared to other adaptive processes, under different scenarios and circumstances.

Data and code availability

Data and scripts used for analysis are available in the Zenodo repository doi.org/10.5281/zenodo.10039728 and doi.org/10.5281/zenodo.10039712

Author contributions

Conceptualization: GMD, JLT, JP, WF, PE; Methodology: GMD, JLT, JP, WF, PE; Investigation: GMD, PHR, GSM, PE; Software: GMD; Data curation: GMD; Formal analysis: GMD, PE; Visualization: GMD; Funding acquisition: PE, JLT, JP; Resources: PE, GSM; Project administration: PE; Supervision: PE; Writing – original draft: GMD, PE; Writing – review & editing: GMD, PHR, GSM, JP, JLT, WF, PE.

Funding

Spanish Ministry of Universities grant FPU17/06268 (GMD), Spanish Ministry of Science and Innovation PID2019-108971GB-I00 (PE), Spanish Ministry of Economy & Competitiveness grant CGL2013-49460-EXP (PE).

References

- Adkins-Regan, E., & Krakauer, A. (2000). Removal of adult males from the rearing environment increases preference for same-sex partners in the zebra finch. *Animal Behaviour*, 60(1), 47–53. https://doi.org/10.1006/anbe.2000.1448
- Akcali, C. K., & Porter, C. K. (2017). Comment on Van Belleghem et al. 2016: Habitat choice mechanisms in speciation and other forms of diversification. *Evolution*, 71(11), 2754–2761. https://doi.org/10.1111/evo.13375
- Bates, D., Mächler, M., Bolker, B., & Walker, S. (2015). Fitting Linear Mixed-Effects Models Using Ime4. *Journal of Statistical Software*, 67, 1–48. https://doi.org/10.18637/jss.v067.i01
- Benkman, C. W. (2017). Matching habitat choice in nomadic crossbills appears most pronounced when food is most limiting. *Evolution*, 71(3), 778–785. https://doi.org/10.1111/evo.13146
- Berner, D., & Thibert-Plante, X. (2015). How mechanisms of habitat preference evolve and promote divergence with gene flow. *Journal of Evolutionary Biology*, 28(9), 1641–1655. https://doi.org/10.1111/jeb.12683
- Birkhead, T. R., Clarkson, K., & Zann, R. (1988). Extra-pair courtship, copulation and mate guarding in wild zebra finches *Taeniopygia guttata*. *Animal Behaviour*, 36(6), 1853–1855. https://doi.org/10.1016/S0003-3472(88)80133-7
- Bolnick, D. I., & Otto, S. P. (2013). The magnitude of local adaptation under genotypedependent dispersal. *Ecology and Evolution*, 3(14), 4722–4735. https://doi.org/10.1002/ece3.850
- Camacho, C., & Hendry, A. P. (2020). Matching habitat choice: It's not for everyone. *Oikos*, 129(5), 689–699. https://doi.org/10.1111/oik.06932
- Camacho, C., Sanabria-Fernández, A., Baños-Villalba, A., & Edelaar, P. (2020). Experimental evidence that matching habitat choice drives local adaptation in a wild population. *Proceedings of the Royal Society B: Biological Sciences*, 287(1927), 20200721. https://doi.org/10.1098/rspb.2020.0721
- Edelaar, P., Baños-Villalba, A., Quevedo, D. P., Escudero, G., Bolnick, D. I., & Jordán-Andrade, A. (2019). Biased movement drives local cryptic coloration on distinct urban pavements. *Proceedings of the Royal Society B: Biological Sciences*, 286(1912), 20191343. https://doi.org/10.1098/rspb.2019.1343

- Edelaar, P., & Bolnick, D. I. (2012). Non-random gene flow: An underappreciated force in evolution and ecology. *Trends in Ecology & Evolution*, 27(12), 659–665. https://doi.org/10.1016/j.tree.2012.07.009
- Edelaar, P., & Bolnick, D. I. (2019). Appreciating the multiple processes increasing individual or population fitness. *Trends in Ecology & Evolution*, 34(5), 435–446. https://doi.org/10.1016/j.tree.2019.02.001
- Edelaar, P., Jovani, R., & Gomez-Mestre, I. (2017). Should I change or should I go? Phenotypic plasticity and matching habitat choice in the adaptation to environmental heterogeneity. *The American Naturalist*, 190(4), 506–520. https://doi.org/10.1086/693345
- Edelaar, P., Otsuka, J., & Luque, V. J. (2023). A generalised approach to the study and understanding of adaptive evolution. *Biological Reviews*, 98(1), 352–375. https://doi.org/10.1111/brv.12910
- Edelaar, P., Siepielski, A. M., & Clobert, J. (2008). Matching habitat choice causes directed gene flow: a neglected dimension in evolution and ecology. *Evolution*, 62(10), 2462–2472. https://doi.org/10.1111/j.1558-5646.2008.00459.x
- Faust, K. M., & Goldstein, M. H. (2021). The role of personality traits in pair bond formation: Pairing is influenced by the trait of exploration. *Behaviour*, 158(6), 447–478. https://doi.org/10.1163/1568539x-bja10076
- Forstmeier, W., Schielzeth, H., Schneider, M., & Kempenaers, B. (2007). Development of polymorphic microsatellite markers for the zebra finch (*Taeniopygia guttata*). *Molecular Ecology Notes*, 7(6), 1026–1028. https://doi.org/10.1111/j.1471-8286.2007.01762.x
- Forstmeier, W., Segelbacher, G., Mueller, J. C., & Kempenaers, B. (2007). Genetic variation and differentiation in captive and wild zebra finches (*Taeniopygia* guttata). Molecular Ecology, 16(19), 4039–4050. https://doi.org/10.1111/j.1365-294X.2007.03444.x
- Garant, D. (2020). Natural and human-induced environmental changes and their effects on adaptive potential of wild animal populations. *Evolutionary Applications*, 13(6), 1117–1127. https://doi.org/10.1111/eva.12928
- Griffith, S. C., Stewart, I. R. K., Dawson, D. A., Owens, I. P. F., & Burke, T. (1999). Contrasting levels of extra-pair paternity in mainland and island populations of the house sparrow (*Passer domesticus*): Is there an 'island effect'? *Biological*

Journal of the Linnean Society, 68(1–2), 303–316. https://doi.org/10.1111/j.1095-8312.1999.tb01171.x

- Holtmann, B., Santos, E. S. A., Lara, C. E., & Nakagawa, S. (2017). Personality-matching habitat choice, rather than behavioural plasticity, is a likely driver of a phenotype– environment covariance. *Proceedings of the Royal Society B: Biological Sciences*, 284(1864), 20170943. https://doi.org/10.1098/rspb.2017.0943
- Johnson, C. N., Balmford, A., Brook, B. W., Buettel, J. C., Galetti, M., Guangchun, L., & Wilmshurst, J. M. (2017). Biodiversity losses and conservation responses in the Anthropocene. *Science (New York, N.Y.)*, 356(6335), 270–275. https://doi.org/10.1126/science.aam9317
- Lüdecke D (2022). sjPlot: Data visualization for statistics in social science. R package version 2.8.10.1, https://CRAN.R-project.org/package=sjPlot.
- Maldonado-Chaparro, A. A., Forstmeier, W., & Farine, D. R. (2021). Relationship quality underpins pair bond formation and subsequent reproductive performance. *Animal Behaviour*, 182, 43–58. https://doi.org/10.1016/j.anbehav.2021.09.009
- Marshall TC, Slate J, Kruuk LEB, Pemberton JM (1998) Statistical confidence for likelihood-based paternity inference in natural populations. *Molecular Ecology*, 7, 639–655.
- Mortier, F., Jacob, S., Vandegehuchte, M. L., & Bonte, D. (2019). Habitat choice stabilizes metapopulation dynamics by enabling ecological specialization. *Oikos*, 128(4), 529–539. https://doi.org/10.1111/oik.05885
- Nicolaus, M., & Edelaar, P. (2018). Comparing the consequences of natural selection, adaptive phenotypic plasticity, and matching habitat choice for phenotype– environment matching, population genetic structure, and reproductive isolation in meta-populations. *Ecology and Evolution*, 8(8), 3815–3827. https://doi.org/10.1002/ece3.3816
- Otto, S. P. (2018). Adaptation, speciation and extinction in the Anthropocene. *Proceedings of the Royal Society B: Biological Sciences*, 285(1891), 20182047. https://doi.org/10.1098/rspb.2018.2047
- Pogány, Á., Vincze, E., Szurovecz, Z., Kosztolányi, A., Barta, Z., Székely, T., & Riebel,
 K. (2018). Personality assortative female mating preferences in a songbird. *Behaviour*, 155(6), 481–503. https://doi.org/10.1163/1568539X-00003500

- Porter, C. K., & Akcali, C. K. (2020). Evolutionary implications of habitat choice. In *eLS* (pp. 85–93). John Wiley & Sons, Ltd. https://doi.org/10.1002/9780470015902.a0029011
- Porter, C. K., & Benkman, C. W. (2022). Performance trade-offs and resource availability drive variation in reproductive isolation between sympatrically diverging crossbills. *The American Naturalist*, 199(3), 362–379. https://doi.org/10.1086/718235
- Pyšek, P., Hulme, P. E., Simberloff, D., Bacher, S., Blackburn, T. M., Carlton, J. T., Dawson, W., Essl, F., Foxcroft, L. C., Genovesi, P., Jeschke, J. M., Kühn, I., Liebhold, A. M., Mandrak, N. E., Meyerson, L. A., Pauchard, A., Pergl, J., Roy, H. E., Seebens, H., ... Richardson, D. M. (2020). Scientists' warning on invasive alien species. *Biological Reviews*, 95(6), 1511–1534. https://doi.org/10.1111/brv.12627
- R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.Rproject.org/.
- Ravigné, V., Dieckmann, U., & Olivieri, I. (2009). Live where you thrive: Joint evolution of habitat choice and local adaptation facilitates specialization and promotes diversity. *The American Naturalist*, 174(4), E141-169. https://doi.org/10.1086/605369
- Regan, C. E., Beck, K. B., McMahon, K., Crofts, S., Firth, J. A., & Sheldon, B. C. (2022). Social phenotype-dependent selection of social environment in wild great and blue tits: An experimental study. *Proceedings of the Royal Society B: Biological Sciences*, 289(1986), 20221602. https://doi.org/10.1098/rspb.2022.1602
- Rice, W. R., & Hostert, E. E. (1993). Laboratory experiments on speciation: What have we learned in 40 years? *Evolution*, 47(6), 1637–1653. https://doi.org/10.1111/j.1558-5646.1993.tb01257.x
- Richardson, J. L., Urban, M. C., Bolnick, D. I., & Skelly, D. K. (2014). Microgeographic adaptation and the spatial scale of evolution. *Trends in Ecology & Evolution*, 29(3), 165–176. https://doi.org/10.1016/j.tree.2014.01.002
- Scheiner, S. M., Barfield, M., & Holt, R. D. (2022). Do I build or do I move? Adaptation by habitat construction versus habitat choice*. *Evolution*, 76(3), 414–428. https://doi.org/10.1111/evo.14355

- Schlichting, C. (2021). *Plasticity and evolutionary theory: where we are and where we should be going* (pp. 367–394). https://doi.org/10.1201/9780429343001-19
- Schluter, D. (2001). Ecology and the origin of species. *Trends in Ecology & Evolution*, *16*(7), 372–380. https://doi.org/10.1016/S0169-5347(01)02198-X
- Schuett, W., Godin, J.-G. J., & Dall, S. R. X. (2011). Do female zebra finches, *Taeniopygia guttata*, choose their mates based on their 'personality'?: Personality and Mate Choice. *Ethology*, 117(10), 908–917. https://doi.org/10.1111/j.1439-0310.2011.01945.x
- Sih, A., Ferrari, M. C. O., & Harris, D. J. (2011). Evolution and behavioural responses to human-induced rapid environmental change. *Evolutionary Applications*, 4(2), 367–387. https://doi.org/10.1111/j.1752-4571.2010.00166.x
- Snowberg, L. K., & Benkman, C. W. (2009). Mate choice based on a key ecological performance trait. *Journal of Evolutionary Biology*, 22(4), 762–769. https://doi.org/10.1111/j.1420-9101.2009.01699.x
- Sultan, S. E., Moczek, A. P., & Walsh, D. (2022). Bridging the explanatory gaps: What can we learn from a biological agency perspective? *BioEssays: News and Reviews in Molecular, Cellular and Developmental Biology*, 44(1), e2100185. https://doi.org/10.1002/bies.202100185
- Trappes, R., Nematipour, B., Kaiser, M. I., Krohs, U., van Benthem, K. J., Ernst, U. R., Gadau, J., Korsten, P., Kurtz, J., Schielzeth, H., Schmoll, T., & Takola, E. (2022). How Individualized Niches Arise: Defining Mechanisms of Niche Construction, Niche Choice, and Niche Conformance. *Bioscience*, 72(6), 538–548. https://doi.org/10.1093/biosci/biac023
- Wadgymar, S. M., DeMarche, M. L., Josephs, E. B., Sheth, S. N., & Anderson, J. T. (2022). Local adaptation: causal agents of selection and adaptive trait divergence. *Annual Review of Ecology, Evolution, and Systematics*, 53(1), 87–111. https://doi.org/10.1146/annurev-ecolsys-012722-035231
- Wang, D., Forstmeier, W., & Kempenaers, B. (2017). No mutual mate choice for quality in zebra finches: Time to question a widely held assumption. *Evolution*, 71(11), 2661–2676. https://doi.org/10.1111/evo.13341
- Waters, C. N., Zalasiewicz, J., Summerhayes, C., Barnosky, A. D., Poirier, C., Gałuszka,
 A., Cearreta, A., Edgeworth, M., Ellis, E. C., Ellis, M., Jeandel, C., Leinfelder,
 R., McNeill, J. R., Richter, D. deB., Steffen, W., Syvitski, J., Vidas, D., Wagreich,
 M., Williams, M., ... Wolfe, A. P. (2016). The Anthropocene is functionally and

stratigraphically distinct from the Holocene. *Science*, *351*(6269), aad2622. https://doi.org/10.1126/science.aad2622

Zann, R. A., Morton, S. R., Jones, K. R., & Burley, N. T. (1995). The timing of breeding by zebra finches in relation to rainfall in Central Australia. *Emu - Austral Ornithology*, 95(3), 208–222. https://doi.org/10.1071/MU9950208

Experimental rapid and small-scale ecological population divergence in the absence of current natural selection

Gabriel Munar-Delgado, Paula Hidalgo-Rodríguez, Gregorio Sánchez-Montes, Jaime Potti, José L. Tella, Wolfgang Forstmeier, Pim Edelaar

Supplementary material

Supplementary methods

Housing and acclimatation phase

For experiment 1, males and females (N = 70) were obtained from different commercial providers to ensure that there were no pre-established pairs. Their age was unknown and there was a variety in plumage colour phenotypes (fully white individuals were avoided since they were hard to sex). Upon arrival in the lab, birds were uniquely colour-banded Males and females were kept visually isolated in two different rooms prior to the experiment for twelve months. For experiment 2 we used young birds only. These birds (N = 103) were bred by ourselves and were the offspring of adults obtained from the same providers as for experiment 1. They were kept with adult individuals until they were 42-57 days old to facilitate sexual imprinting (Vos, 1995).

All birds were fed ad libitum and their diet consisted of a mixture of Prestige Tropical Finches seeds (Versele-Laga), grit, cuttlefish bone, and tap water (with extra vitamins once a week). When birds were not in the breeding experiment, fresh leaves (of wild plants) and commercial egg food were also provided.

Adult male and female birds from experiment 1 were isolated until the beginning of the experiment. Half of the males and half of the females were randomly assigned to group A (access to the set of feeders in area A) and the other half to group B (access to the other set of feeders in area B). We first introduced the females in the experimental aviary and started the acclimation phase. During this phase, we placed 10 electronic feeders evenly distributed in the two centres of areas A and B and programmed them to stay open, so all birds could have access to the seeds. Once we confirmed that all birds were eating, we programmed the feeders to stay closed until a bird with a PIT-tag approached. In this way, birds got used to the opening mechanism. Then, we started to move the feeders throughout the aviary every $2(\pm 1)$ days. We first moved all of them to one breeding area of the aviary and then all to the other area. Thus, we ensured that all individuals explored the entire aviary. In the following step, we reprogrammed the feeders so that half of them only gave access to food to birds of group A, and placed them in area A. The other feeders were programmed to only give access to group B, and were placed in area B. Thus, we generated two novel local environments where individuals had a different local ecological performance based on their transponder type, a novel matching ecological trait. Next, for a week, we let them learn by trial and error where each individual had access to food (i.e., let them assess their local performance across local environments). After this, we moved

the females out of the experimental aviary, introduced the males and repeated the acclimation procedure previously used for the females. At this point we removed from the experiment 14 individuals that had health issues (N = 7) or that were thought to feed on spilled seeds or on "stolen" seeds from feeders in the area where they were supposed to not have access to food (N = 4), or to balance the male:female ratio after removing the previous birds (N = 3) (see below). Thus, 28 males and 28 females stayed in the experiment. After that, we placed 9 feeders in area A and 9 feeders in area B, giving access only to individuals from the corresponding group A or B. Next, we introduced the experienced females with the males in the experimental aviary, and let all individuals move freely. After one week, we placed 28 nest boxes in the experimental aviary, half at each breeding area, and let the individuals breed.

In experiment 2, we repeated the procedures used in experiment 1 but used males and females that sexually matured within the experimental aviary and without any separation between the sexes prior to the experiment (i.e., the individuals already grew up in an ecologically heterogeneous environment). To start the acclimation, electronic feeders were placed at both central aviary areas when birds were 15-30 days old. However, conventional feeders were not removed until the age of 33-48 days, and until the age of 42-57 days a few experienced adults with transponders were present to show the young birds how to use the electronic feeders. After the adults were removed, we first moved all the feeders across the experimental aviary and then separated them between both future breeding areas, as for experiment 1. At this point, 33 birds were removed. They either had health issues (N = 3), were not able to use the feeders (N = 6), were thought to feed on spilled seeds or to "steal" seeds from feeders in the area where they were supposed to not have access to food (N = 5), or were removed to improve the male:female ratio after removing the previous birds (N = 19) (see below). Then, after the remaining individuals had reached sexual maturity (between the age of 92-107 days; males N = 38 and females N = 32), we introduced 34 nest boxes, half at each breeding area, and let the individuals breed.

Removing individuals before the experiment

Some birds were removed from the experimental aviary before starting the experiment either because (i) they had health issues; (ii) they were not able to use the feeders; (iii) they were thought to feed on spilled seeds or "steal" seeds from feeders in the area where they were supposed to not have access to food; (iv) there was an unbalanced number of males and females. More details are provided below to facilitate future studies using a similar approach.

(i) **Birds with health issues.** We received some birds from the suppliers already in poor health. Some were unable to maintain flight for more than a few seconds (an impediment to move between breeding areas, and for breeding in general). We also observed some birds that were affected by the transponder tag. These suffered from a swollen tarsus, which was aggravated by the transponder. In some birds, the swelling was due to infections with scaly leg mites (which were subsequently treated with ivermectin, 0,12%).

In others, it seems that the provided egg food or dirt (including scaly material from the tarsus) was accumulating between the transponder tag and the leg and that was causing some kind of infection. This situation was improved by providing the egg food in closed feeders so birds were not able to perch on the food. Birds in this condition were treated with an extract of *Centella asiatica* in the form of ointment (10 mg/g, Blastoestimulina) and we changed their transponder tag to their other "healthy" leg. Even so, some individuals remained with some swelling. This prevented them from carrying the transponder tag, and for this reason they were removed.

(ii) and (iii) **Birds to which the experimental treatment could not be properly applied.** Some birds were not able to learn how to use the feeders. The reason is unknown but these birds were registered at a very low frequency (compared to their conspecifics) or not registered at all at the electronic feeders. Thus, the experimental treatment, to provide access to feeders (and food) in only one breeding area, could not be applied to them and, moreover, they were at risk of starvation. Because of this they were removed. On the other hand, some birds were registered at a high frequency (compared to their conspecifics) at the electronic feeders where they were supposed not to have access to food. These birds were either unable to learn where they had access to food (because they keep trying to get seeds from the wrong feeders), or somehow, they were able to feed from spilled seeds on those feeders. We improved the latter by placing grids under the feeders to collect the seeds that fell from the feeders when a bird was feeding and thereby prevent other birds from having access to those seeds. In either case, the experimental treatment (giving access to food in only one breeding area) was not successful for those birds, so they were removed.

(iv) **Birds removed to balance the female:male ratio.** In experiment 1, the acclimation phase started with the same number of males and females. After removing some individuals due to points i, ii and iii, 3 additional males were removed to reach a 1:1 sex ratio. In experiment 2, the acclimation phase started with an unequal number of males and females. After removing some individuals due to points i, ii and iii above, 19 additional males were removed. We entered 32 females and 38 males in the experiment. We maintained 6 "spare" males due to the relatively high proportion of female same-sex pairs that bred in experiment 1, to make sure that this wasn't due to the lack of suitable male partners.

Identification of breeders

For camera recordings, we placed the camera attached to a tripod 50 cm away from the target nest box in such a way that it allowed us to see the colour band combination of the individuals entering or leaving the nest box. We then associated breeding individuals with a nest based on individuals' behaviour during feeding episodes after young had hatched. We observed that breeding individuals performed nest guarding: one individual stayed inside the nest box until the other breeding individual perched on top of the nest box, in front of the entrance or entered the nest box. Then the guarding individual came out, and the second individual stayed guarding the nest. Zebra finches often (try to) visit other nest

boxes. In case the visiting individual was not a breeding individual from the nest, the guarding individual displayed aggressive behaviour towards it. Thus, we associated an individual with a nest box and a brood when we observed that a guarding individual let it go inside the nest box without displaying aggressive behaviour. We discarded occasional visits from individuals when there was no guarding individual present. Not all breeding pairs were typical male-female pairs, so we classified the breeding individuals as being part of a heterosexual pair, same-sex pair, or trio (when 3 individuals allowed each other to enter the nest).

PCR conditions

DNA was isolated using an extraction robot (Freedom EVO 100, Tecan). All DNA samples were genotyped at 10 microsatellite loci (Forstmeier et al., 2007, Table S9) with PCR multiplex cycles consisting of initial denaturation (95°C, 5 min), 23-30 cycles (depending on the multiplex reaction) of denaturation (95°C, 30 s), annealing (59°C or 60°C) and extension (72°C, 30 s), followed by a final extension step (60°C, 30 min). All reactions were run in a total volume of 10 μ l, containing 5 μ l of Type-it Master Mix (Qiagen), 1 μ l of primer mix, 3 μ l of H2O and 1 μ l of DNA (Table S9). Genotyping was performed on an ABI PRISM 3130 sequencer with the GeneScan 500 LIZ standard. Allele peaks were assigned manually in GeneMapper ver. 4.0.

Results

Data summary

Table S1. Overview of where zebra finches bred, and with whom. Based on visual observations on breeding individuals (raising fledglings in a nest). See Methods for a description of each classification. 1 There were 49 breeding heterosexual individuals, but one polygamous male and its two female partners were excluded to simplify the analysis

Breeders						Non-breeders
68					58	
	Heterosexual Same-sex/trios					
	4	16 ¹		1	9	
Habitat n	Habitat matching Assortative mating			Habitat r	natching	
Yes	No	Yes	No	Yes	No	
36	10	34	12	14	5	

Table S2. Overview of where female or male zebra finches produced

offspring. Based on genetic parentage analysis. See Methods for a description of each classification.

Reproductive individuals						
Female-nest c	ombinations	Male-nest combinations				
34	1	49				
Matc	hing	Matching				
Yes	No	Yes	No			
25	9	29	20			

Table S3. Model estimates for matching habitat choice for breeding area for heterosexual pairs (visual observations). Binomial GLM (N=23).

Fixed effect	Estimate ± standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	-1.7919±1.1977	-	-	-
Male transponder type	2.4162±1.3193	1	4.0157	0.045
Female transponder type	2.0323±1.1489	1	3.3177	0.069
Experiment	0.2186±1.2863	1	0.0290	0.865

Table S4. Model estimates for matching habitat choice for breeding area for the entire breeding population (visual observations). Binomial GLMM (N=70). Individual ID was fitted as random with an estimated variance component of 0.74 ± 0.86 (SD).

Fixed effect	Estimate \pm standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	-2.1824±1.4630	-	-	-
Transponder type	2.7411±1.7846	1	24.0110	9.580 x 10 ⁻⁷
Sex	0.6301±0.9741	1	2.2913	0.130
Experiment	0.5767±0.7847	1	0.6970	0.404
Transponder type:Sex	-1.0838±1.5351	1	0.51253	0.474

Table S5. Model estimates for reproductive success. Poisson GLM (N=23).

Fixed effect	Estimate \pm standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	0.8521±0.4937	-	-	-
N° of matching individuals	0.1061±0.2354	1	0.2079	0.648
Experiment	-0.2536±0.2957	1	0.72326	0.395

Table S6. Model estimates for matching habitat choice by genetic mothers. Binomial GLMM (N=34). Individual ID was fitted as random with an estimated variance component of 0.22±0.47 (SD).

Fixed effect	Estimate \pm standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	-2.359±1.476	-	-	-
Female transponder type	2.291±1.356	1	8.0714	4.497x 10 ⁻³
Experiment	1.034±1.071	1	1.2576	0.262

Table S7. Model estimates for matching habitat choice by genetic fathers. Binomial GLMM (N=35). Individual ID was fitted as random with an estimated variance component of 1.21±1.10 (SD).

Fixed effect	Estimate \pm standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	-1.6680±1.4630	-	-	-
Male transponder type	0.8860±0.8322	1	1.2248	0.268
Experiment	1.4730±0.9855	1	2.8984	0.089

Table S8. Model estimates for matching habitat choice by genetic fathers including their within-pair/extra-pair condition. Binomial GLMM (N=33). Individual ID was fitted as random with an estimated variance component of 0.19 ± 0.44 (SD).

Fixed effect	Estimate ± standard error	Degrees of freedom	χ2	<i>p</i> -value
Intercept	-1.3162 ± 1.0464	-	-	-
Transponder type	2.7542± 1.4918	1	1.5435	0.214
Extra-pair	1.5334 ± 1.2376	1	0.0266	0.871
Experiment	0.6119 ± 0.9309	1	0.6112	0.434
Male transponder type:Extra-pair	-3.4116 ± 1.9847	1	4.3169	0.038

Genotyping

chromosome	primer name	fluorescence label	mix	mix volume	volume of primer (stock concentration 100µM)	annealing temperature	N of cycles
TgulA	chr1A_39MB	NED	1	300	0.7	60°C	23
Tgu3	chr3_58MB	PET	1	300	1.3	60°C	23
Tgu5	chr5_34MB	PET	3	200	0.8	59°C	30
Тдиб	chr6_16MB	VIC	3	200	6	59°C	30
Tgull	chr11_8MB	NED	3	200	0.5	59°C	30
Tgu14	chr14_9MB	NED	2	200	0.5	60°C	25
Tgu15	chr15_6MB	6FAM	1	300	1.5	60°C	23
Tgu22	chr22_3MB	VIC	1	300	0.7	60°C	23
Tgu26	chr26_3MB	6FAM	2	200	1	60°C	25
Tgu27	chr27_1MB	6FAM	3	200	0.5	59°C	30

Table S9. Primers and PCR conditions for the 10 microsatellite markers used for parentage assignment.

Supplementary references

- Forstmeier, W., Schielzeth, H., Schneider, M., & Kempenaers, B. (2007). Development of polymorphic microsatellite markers for the zebra finch (Taeniopygia guttata). *Molecular Ecology Notes*, 7(6), 1026–1028. https://doi.org/10.1111/j.1471-8286.2007.01762.x
- Vos, D. R. (1995). The role of sexual imprinting for sex recognition in zebra finches: A difference between males and females. *Animal Behaviour*, 50(3), 645–653. https://doi.org/10.1016/0003-3472(95)80126-X